Author’s response to reviews

Title: Effects of cod intake in pregnancy on iodine nutrition and infant development: study protocol for Mommy’s Food - a randomized controlled trial

Authors:

Maria Markhus (maria.wik.markhus@hi.no)
Ingrid Kvestad (ingrid.kvestad@uib.no)
Lisa Midtbø (lisakolden.middbo@hi.no)
Ive Nerhus (ive.nerhus@hi.no)
Elisabeth Ødegård (elisabeth.odegard@hi.no)
Ingvild Graff (ingvild.graff@uni.no)
Øyvind Lie (oyvind.lie@fiskeridir.no)
Lisbeth Dahl (lisbeth.dahl@hi.no)
Mari Hysing (mari.hysing@uni.no)
Marian Kjellevold (marian.kjellevold@hi.no)

Version: 1 Date: 23 Jan 2018

Author’s response to reviews:

Dear Editor,

Thank you for the assessment of the manuscript «Effects of cod intake in pregnancy on iodine nutrition and infant development: study protocol for Mommy’s Food – a randomized controlled trial », in BMC Nutrition as a study protocol.

1. Spirit checklist

A spirit checklist has been uploaded with the revised manuscript

2. Consent to participate
A clear statement of consent has been included in the “ethical approval and consent to participate section: “Written informed consent was obtained from the participants whom can withdraw from the study at any time without giving any reason.”

3. Clean version

A clean version of the manuscript without any tracked changes has been uploaded.

For your info, National Institute of Nutrition and Seafood Research in Bergen, Norway has changed name after fusion with Institute of Marine Research in Bergen, Norway to Institute of Marine Research.

Sincerely yours,

Maria Wik Markhus